Author's response to reviews

Title: Racial disparities in infant mortality: What has birth weight got to do with it and how large is it?

Authors:

Timothy B Gage (tbg97@albany.edu)
Fu Fang (ff2393@albany.edu)
Erin K O'Neill (eo0379@albany.edu)
A G DiRienzo (gdirenzo@albany.edu)

Version: 7 Date: 17 December 2010

Author's response to reviews:

We have corrected a few additional typos.

Table 4 was formated differently in the galley than in the document that we submitted.

We have tried to correct this but it will need to be checked, since we can not see the problem in our document.